Author’s response to reviews

Title: Anti-inflammatory activity of Jefea gnaphalioides (A. Gray), Astereaceae.

Authors:
Axel Villagómez (axelvillagomezrodriguez@gmail.com)
Julia Pérez (jperez@correo.xoc.uam.mx)
Ana Laura Esquivel (aesquivel@correo.xoc.uam.mx)
Cuauhtemoc Pérez (cperezg@correo.xoc.uam.mx)
Claudia Angélica Soto (casoto@correo.xoc.uam.mx)
Salud perez (msperez46@yahoo.com.mx)

Version: 3 Date: 10 Jul 2019

Author’s response to reviews:

July 10th, 2019.

Dear Liam Messin
BMC Complementary and Alternative Medicine

We appreciate the comments and suggestions of reviewer 2 and editor, to improve the quality of the manuscript BCAM-D-17-01523R1 Anti-inflammatory activity of Jefea gnaphalioides (A. Gray), Astereaceae. We did the next additional assays

1) We determined total phenolic content by using Folin Ciocalteu assay.
2) We deleted Shinoda and Marini-Bettolo assays, they were change for the test of flavonoids content by AlCl3 complexation method.

Also, we did the next changes:
1) The anti-inflammatory activity of palmitic acid was eliminated.
2) The results of total phenolic and flavonoid content were included
3) Discussion section was improved with the additional results.
4) Five new references were included.
5) Two erroneous references were corrected.
6) The citations in the text have been thoroughly revised.
All changes and additional methods are in red.

The English language was reviewed again, by someone who is fluent in English.

We are sending you the new certificate of English for American Journal Experts, who review the manuscript.

We hope that with these corrections the manuscript will be suitable to be published in your prestigious journal.

Sincerely Yours

Dr. Salud Pérez